

Format for ANSWERING REVIEWERS



October 1, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: Revised manuscript12367.docx).

Title: Unraveling the role of EGFR in oral lesions: Key to non surgical treatment modes

Author: Bhavna C Mohan,Punnya V Angadi

Name of Journal: *World Journal of Stomatology*

ESPS Manuscript NO: 12367

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

1. This review is well-planned and seems satisfying. However, abstract needs re-organization. It is publishable when abstract is organised in a more explanatory manner.

The abstract has been modified and organised as suggested by the reviewer

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Stomatology*.

Sincerely yours,

A handwritten signature in blue ink, appearing to read 'Punnya V. Angadi', with a horizontal line underneath the name.

Dr. Punnya. V. Angadi, MDS, DNB

Reader, Department of Oral Pathology and Medicine

KLEVK Institute of Dental Sciences and Hospital,

Belgaum-590010,Karnataka- INDIA.

Telephone No. – 9880845530

Fax : (0831) -2470640

E-mail: punnya_angadi@rediffmail.com